ORIGINAL RESEARCH



Adherence to Triple Single-Pill Combination of Perindopril/Indapamide/Amlodipine: Findings from Real-World Analysis in Italy

Claudio Borghi · Pathiyil Balagopalan Jayagopal · Alexandra Konradi · Luiz Aparecido Bortolotto · Luca Degli Esposti · Valentina Perrone · Jacques R. Snyman

Received: November 30, 2022 / Accepted: February 1, 2023 / Published online: February 24, 2023 \odot The Author(s) 2023

ABSTRACT

Introduction: Single-pill combination therapy for hypertension is recognized to improve adherence to treatment. However, less is known about the benefits of triple single-pill combinations. This retrospective observational analysis aimed to assess changes in adherence when treatment was switched from perindopril (PER)/ indapamide (IND) + amlodipine (AML) to PER/ IND/AML single-pill combination, in Italian clinical practice.

C. Borghi (🖂)

University of Bologna, IRCCS Ospedale S. Orsola, Bologna, Italy e-mail: claudio.borghi@unibo.it

P. B. Jayagopal Lakshmi Hospital, Palakkad, India

A. Konradi Almazov National Medical Research Center, St. Petersburg, Russian Federation

L. A. Bortolotto Instituto do Coração, Hospital das Clinicas-FMUSP, Sao Paolo, Brazil

L. Degli Esposti · V. Perrone CliCon S.r.l, Società Benefit-Health, Economics and Outcomes Research, Bologna, Italy

J. R. Snyman Forte Research (Pty Ltd) and Private Practice, Pretoria, South Africa *Methods*: This analysis used data extracted from administrative databases of Italian healthcare entities. Adult patients receiving PER/IND/AML were selected, and the prescription date was considered as the index date. Among them, those who had a prescription for PER/IND + AML during the 12 months before the index date and a prescription of PER/IND/ AML during 6 months of follow-up were included. Adherence was calculated as the proportion of days covered (PDC: PDC < 40%, nonadherent; PDC = 40–79%, partially adherent; PDC \geq 80%, adherent).

Results: Among the identified patients, 158 were exposed users and were included in the analysis. When patients were compared before and after switch to triple single-pill combination, the proportion of adherent patients was significantly higher with PER/IND/AML single-pill combination (75.3%) than with PER/IND + AML combination (44.3%) (P < 0.05). Conversely, the proportion of non-adherent patients was lower with the PER/IND/AML single-pill combination (14.6%) vs PER/IND + AML (17.7%) (P < 0.001).

Conclusion: This real-world analysis showed that switching to a triple single-pill combination could offer an opportunity to improve adherence to antihypertensive treatment in real-life clinical practice.

Adv Ther (2023) 40:1765-1772

PLAIN LANGUAGE SUMMARY

Medication adherence is defined by the World Health Organization as the "extent to which a person's behavior (in taking medication) corresponds with agreed recommendations from a healthcare provider". Low levels of medication adherence in hypertension have been linked with increased disease burden and with higher costs for patients. Patients with hypertension whose blood pressure is poorly controlled often need to receive more than one pill. Nevertheless, having to take many pills may result in poor adherence, i.e., patients not taking their treatment as prescribed. Combining multiple drugs into a single pill for the management of hypertension is known to improve adherence; however, limited evidence exists about the benefits of triple single-pill combinations compared with equivalent free combinations in real clinical practice. This analysis evaluated changes in adherence before and after patients switched from a three-drug therapy of perindopril/indapamide single-pill + amlodipine (PER/IND + AML) to perindopril/indapamide/amlodipine (PER/IND/AML) taken as a single pill. In this analysis, real-world data from Italian administrative databases covering around 11% of the Italian population were used. Overall, 158 patients were included. More patients were found to be adherent after switch to PER/IND/AML single pill (75.3% vs 44.3% of PER/IND + AML combination). Partially adherent and poorly adherent patients were fewer with PER/IND/AML single-pill combination (10.1% and 14.6%, respectively) compared to PER/IND + AML combination (38.0% and 17.7%, respectively). These findings indicate that switching to a simplified therapy in which all three drugs are taken in one pill may offer an opportunity for increasing the number of patients that are adherent to their medication.

Keywords: Hypertension; Adherence; Perindopril; Indapamide; Amlodipine; Triple single-pill combination therapy

Key Summary Points

Why carry out this study?

Many patients with hypertension do not attain blood pressure control on two antihypertensive agents and require additional medication.

To reduce the pill burden and improve adherence, triple single-pill combinations are available, but data on their benefits compared with equivalent free combinations outside the clinical trial setting are limited.

This real-world analysis used data from Italian administrative health databases to evaluate changes in adherence when treatment was switched from a three-drug therapy comprising perindopril/ indapamide single-pill + amlodipine (PER/IND + AML) to a PER/IND/AML single-pill formulation.

What was learned from the study?

The proportion of fully adherent patients was significantly higher with the PER/ IND/AML single-pill compared with the PER/IND + AML combination.

Simplifying the antihypertensive treatment regimen by using a triple singlepill combination may offer an opportunity to improve adherence in reallife clinical practice.

INTRODUCTION

The control of cardiovascular (CV) risk factors, particularly arterial hypertension, and the achievement of the related therapeutic goals are key factors in the treatment of CV conditions. Poor control of hypertension leads to progressive target organ damage and impacts on CV mortality. Real-world studies indicate that, in Italy, only 30–40% of patients undergoing

pharmacological treatment are adequately controlled [1]. Suboptimal adherence to antihypertensive treatment is an important cause of poor blood pressure control [2]. Moreover, poor adherence is associated with increased risk of CV events, comorbidities and mortality, and increased healthcare resource consumption and related costs [3]. Therefore, efforts directed at improving adherence should be recognized as an integral part of hypertension management. Pill burden may affect medication adherence, as it has been reported that increasing the number of antihypertensive medications is associated with a higher rate of nonadherence [3]. In this context, we have conducted a real-world analvsis to evaluate changes in adherence when treatment was switched from perindopril/indapamide single-pill + amlodipine (PER/IND + AML) to PER/IND/AML single-pill formulation, in Italian clinical practice.

METHODS

This observational retrospective analysis used the administrative databases (demographic, pharmaceutical, hospitalization, and outpatient service databases) of a sample of local health units (LHU) covering around 11% of the Italian population. These databases store all data concerning the healthcare resources reimbursed by the Italian National Health Service (INHS). All adult patients (at least 18 years of age) treated with PER/IND/AML as single-pill combination during the study period (2010 to 2020) were screened for eligibility in the study. The date of first detection for this prescription was defined as the index date. Exposed users, defined as those receiving PER/IND + AML combination during 12 months before the index date, were included in the study (Fig. 1). All these patients had data available from at least 1 year before the period) date (characterization index to 6 months afterwards (follow-up). Patients were excluded if they moved to another region or died during the 6 months of follow-up. Adherence was evaluated as proportion of days covered (PDC), i.e., the ratio between the number of days of medication supplied and days of observation, multiplied by 100, by using the

following cutoffs: PDC < 40% (non-adherent); PDC = 40-79%(partially adherent): PDC > 80% (adherent). PDC is considered as one of the most reliable methods to measure adherence in chronic therapies, and these cutoffs are widely used in the literature to evaluate levels of adherence [4, 5]. Adherence to PER/IND + AML combination was calculated as the number of days covered by all three drugs (in two tablets) in the 6 months prior to the index date, and adherence to PER/IND/AML single-pill combination was calculated as the days covered by single-pill combination in the 6 months after the index date. Statistical significance was defined as P < 0.05. All analyses were performed using Stata SE version 12.0 (StataCorp, College Station, TX, USA).

Ethics Statement

To guarantee patients' privacy, an anonymous univocal numerical code was assigned to each subject included in the study, in full compliance with the European General Data Protection Regulation (GDPR) (2016/679). This code allowed the electronic linkage between all different databases. No identifiers related to patients were provided to the authors. All the results of the analyses were produced as aggregated summaries, which are not possible to assign, either directly or indirectly, to individual patients. Informed consent was not required (pronouncement of the Data Privacy Guarantor Authority, General Authorization for personal data treatment for scientific research purposes no. 9/2014), and the LHU ethics committees approved the study.

RESULTS

In this analysis, of the 50,679 patients with PER/ IND/AML single-pill formulation identified, only 158 were exposed users and were included in the analysis. Exposed users were defined as those receiving a prescription for PER/IND + AML during the 12 months before the index date (i.e., PER/IND/AML single-pill prescription date and 6 months of follow-up period with PER/IND/AML. Mean age was 67.9 years, and



$Identification \ of \ patients \ moving \ from \ PER/IND+AML \ to \ PER/IND/AML \ single-pill \ combination.$

Fig. 1 Study design

most patients (65.8%) were in the age range of 60–79 years; male patients accounted for 63.9%. Other antihypertensive treatments were previously prescribed to 59.5% of patients, and lipid-lowering agents to 55.7%, most of which were statins.

The proportion of adherent patients was significantly higher with PER/IND/AML singlepill combination than with PER/IND + AML combination (75.3% vs 44.3%, *P* value < 0.05) (Fig. 2). Partially adherent patients were detected more often while on PER/IND + AML (38.0%) than on single-pill combination (10.1%) (*P* < 0.001), while the proportion of poorly adherent patients was lower with the single-pill combination (14.6%) vs PER/IND + AML (17.7%) (P < 0.001).

DISCUSSION

Suboptimal adherence strongly contributes to poor blood pressure control [6, 7]. However, in the real word, in ambulatory patients with multiple diseases who are often treated with multiple drugs, suboptimal adherence is still highly prevalent [8], and only half of patients with hypertension persist with their treatment after 1 year [7]. It has been reported that biochemically confirmed non-adherence increases with the number of drugs taken, and in one study of 1348 patients with hypertension from





AML single-pill combination was identified by using Anatomical Therapeutic Chemical (ATC) codes C09BA04b + C08CA01, and C09BX01, respectively two European countries, non-adherence was detected in 79.3% of patients treated with six drugs or more [8].

Hence, regimen simplification represents one of the strategies to pursue the improvement of adherence to antihypertensive medication. A recent meta-analysis of studies that measured adherence to antihypertensive medication (including randomized controlled trials, observational studies, and retrospective database analyses) reported that in 78% of studies, adherence significantly improved in patients receiving single-pill combinations versus free equivalent combinations [2]. Furthermore, 88% of the studies showed that patients receiving single-pill combinations had significantly improved persistence or were significantly less likely to discontinue therapy than patients receiving a free combination. There were also significantly greater reductions in blood pressure with the single-pill combination, suggesting that the improved adherence associated with the simplified regimen had led to better blood pressure control.

Adherence in clinical practice can differ markedly from the clinical trial setting and retrospective analyses are an accepted way to assess adherence in routine clinical practice [9]. Selecting an optimal antihypertensive regimen can be difficult given the multiple options available. It is therefore important to examine the relative clinical effectiveness of different drug classes and their combinations in the realworld setting, particularly as patient profiles often differ from those included in trials. The wide accessibility of healthcare utilization databases in electronic format, such as used in the current study, has greatly simplified this process and confirmed the benefits of single-pill combination therapy for improving adherence [9].

Consequently, the World Health Organization (WHO) recommends single-pill combinations as the emerging best practice for safe, effective, rapid, and convenient hypertension control, and single-pill combination antihypertensive medications have been added to the WHO essential medicines list [10]. Additionally, all recent guidelines place strong emphasis on the issue of non-adherence, notably by recommending single-pill combination therapy as the preferred strategy for initial two-drug combination treatment and the preferred use of single-pill combination therapy for most patients [11, 12]. In addition, it was observed that patients treated with three antihypertensive agents or more generally have more associated comorbidities and advanced disease [13]. Thus, their pill burden will be high, making adherence to all prescribed treatments even more challenging. The better adherence associated with single-pill combination therapy might be of particular interest in the case of polypharmacy regimens, e.g., in those presenting with several comorbid conditions requiring different associated treatments and in the elderly [2, 14]. This group of patients (elderly or with a multimorbid profile) are generally underrepresented in randomized clinical trials conducted worldwide.

However, data on triple single-pill combination treatment are limited, with a lack of direct comparisons between regimens and on the potential benefits of triple single-pill combinations versus the same free combination. The efficacy of the switch to the triple PER/IND/ AML single-pill combination in patients with hypertension was demonstrated in the realworld PETRA study [15].

A recent observational study conducted in a real-life clinical setting found high adherence (a primary endpoint of this study) to treatment with the PER/IND/AML single-pill combination [16]. The simplification of the antihypertensive drug regimen by administration of one instead of three different pills, the rapid achievement of target blood pressure levels, and the low incidence of adverse events may explain the high levels of adherence observed. Similarly, in patients with grade 2-3 hypertension and high or very high cardiovascular risk who did not reach blood pressure target when using the free combination of PER, IND, and AML, there was an improvement in adherence when single-pill combination therapy was used.

Our results, reporting a significantly higher proportion of adherent patients after switching from a multiple-pill to a triple single-pill combination of PER/IND/AML, are in line with these findings. The main limitation of the analysis is the small number of exposed patients included. A further limitation is the lack of data on factors that could have affected the level of medication adherence (i.e., social, economic, or clinical factors) not reported in the administrative database, as well as data on blood pressure level, which are not present within the databases. Furthermore, to calculate adherence, we have assumed that a patient takes one pill per day, as it is not possible to retrieve information on the actual intake of the patient from administrative databases. A strength of this study is the unselected population represented in real-life settings.

CONCLUSION

Switching to a triple single-pill combination could offer an opportunity to improve adherence to antihypertensive treatment in real-life clinical practice.

ACKNOWLEDGEMENTS

Funding. Open access funding provided by Alma Mater Studiorum - Università di Bologna within the CRUI-CARE Agreement. The study report was developed by Clicon S.r.l. Società Benefit and funded by Servier, France. The journal's Rapid Service and Open Access Fees was funded by Servier, France.

Medical Writing and Editorial Assistance. Medical writing assistance was provided by Melania Dovizio of Clicon S.r.l. Società Benefit and funded by Servier, France.

Authorship. All named authors meet the International Committee of Medical Journal Editors (ICJME) criteria for authorship for this article, take responsibility for the integrity of the work as a whole, and have given their approval for this version to be published.

Author Contributions. All authors interpreted the data, provided critical feedback on

the manuscript, approved the final manuscript for submission, and are accountable for the accuracy and integrity of the manuscript.

Disclosures. Claudio Borghi has received consultation fees and lecture honoraria from Servier, Novartis, Menarini Corporate, Novo Nordisk, Alfasigma, Sanofi. Pathiyil Balagopalan Jayagopal declares no conflict of interest. Alexandra Konradi: lecture honoraria from Servier, Novartis, Novo Nordisk, KRKA. Luiz Aparecido Bortolotto declares Advisory board member for Servier, Medtronics and Merck. Jacques R. Snyman has received consultation fees and lecture honoraria from Servier, Novartis, Novo Nordisk, Boehringer Ingelheim, Merck, AbbVie. Valentina Perrone and Luca Degli Esposti declare no conflict of interest.

Compliance with Ethics Guidelines. Informed consent was not required (pronouncement of the Data Privacy Guarantor Authority, General Authorization for personal data treatment for scientific research purposes n.9/2014), and the LHU ethics committees approved the study: Comitato etico interprovinciale Area I (protocol number 63/CE/20 and 68/CE/20, approval date 3/12/2020); Comitato Etico per le Sperimentazioni Cliniche (CESC) della Provincia di Vicenza (protocol number 1627, approval date 28/10/2020); Comitato Etico "Lazio 2" (protocol number 0179046/2020, approval date 28/10/2020); Comitato Etico Inter-aziendale Campania Sud (protocol number 51, approval date 02/09/2020 and protocol number 64, approval date 03/11/ 2020); Comitato Etico "Lazio 1" (protocol number 1166/CE Lazio 1, approval date 12/10/ 2020 and protocol number 1079/CE Lazio 1, approval date 23/09/2020); Comitato Etico per la Sperimentazione Clinica della provincia di Venezia e IRCCS S.Camillo (approval date 28/07/2020); Comitato Etico per le province di L'Aquila e Teramo (protocol number 11, approval date 24/03/2021); Comitato Etico Regionale Umbria (protocol number 19414/20/ ON, approval date 16/09/2020); Autorizzazione CE Imperia Comitato Etico Regionale della Regione Liguria (protocol number 024/2019, approval date 17/06/2019). This study was

1771

performed in accordance with the Helsinki Declaration of 1964 and its later amendments.

Data Availability. The data sets generated and/or analyzed for this study are not publicly available as they include medical records of patients from a secondary source.

Open Access. This article is licensed under a Creative Commons Attribution-NonCommercial 4.0 International License, which permits any non-commercial use, sharing, adaptation, distribution and reproduction in any medium or format, as long as you give appropriate credit to the original author(s) and the source, provide a link to the Creative Commons licence, and indicate if changes were made. The images or other third party material in this article are included in the article's Creative Commons licence, unless indicated otherwise in a credit line to the material. If material is not included in the article's Creative Commons licence and your intended use is not permitted by statutory regulation or exceeds the permitted use, you will need to obtain permission directly from the copyright holder. To view a copy of this licence, visit http://creativecommons.org/licenses/bync/4.0/.

REFERENCES

- 1. Borghi C, Desideri G, Tocci G, Trimarco B, Nati G. Aderenza alla terapia delle malattie cardiovascolari croniche: nuove soluzioni. G Ital Cardiol. 2021;22(5 Suppl 1):e86–91.
- Parati G, Kjeldsen S, Coca A, Cushman WC, Wang J. Adherence to single-pill versus free-equivalent combination therapy in hypertension: a systematic review and meta-analysis. Hypertension. 2021;77(2):692–705.
- 3. Weisser B, Predel HG, Gillessen A, et al. Single pill regimen leads to better adherence and clinical outcome in daily practice in patients suffering from hypertension and/or dyslipidemia: results of a meta-analysis. High Blood Press Cardiovasc Prev. 2020;27:157–64.
- 4. Degli Esposti L, Perrone V, Veronesi C, et al. Modifications in drug adherence after switch to fixeddose combination of perindopril/amlodipine in

clinical practice. Results of a large-scale Italian experience. The amlodipine-perindopril in real settings (AMPERES) study. Curr Med Res Opin. 2018;34:1571–7.

- Raebel MA, Schmittdiel J, Karter AJ, Konieczny JL, Steiner JF. Standardizing terminology and definitions of medication adherence and persistence in research employing electronic databases. Med Care. 2013;51(8 Suppl 3):S11–21.
- Vrijens B, Vincze G, Kristanto P, Urquhart J, Burnier M. Adherence to prescribed antihypertensive drug treatments: longitudinal study of electronically compiled dosing histories. BMJ. 2008;336(7653): 1114–7.
- 7. Burnier M, Egan BM. Adherence in hypertension. Circ Res. 2019;124(7):1124–40.
- 8. Gupta P, Patel P, Štrauch B, et al. Risk factors for nonadherence to antihypertensive treatment. Hypertension. 2017;69(6):1113–20.
- 9. Corrao G, Mancia G. Research strategies in treatment of hypertension: value of retrospective reallife data. Eur Heart J. 2022;43(35):3312–22.
- 10. World Health Organization. Executive summary: the selection and use of essential medicines 2019: report of the 22nd WHO Expert Committee on the selection and use of essential medicines: WHO Headquarters, Geneva, 1–5 April 2019. World Health Organization. 2019. https://apps.who.int/iris/handle/10665/325773.
- 11. Williams B, Mancia G, Spiering W, et al. 2018 ESC/ ESH Guidelines for the management of arterial hypertension. Eur Heart J. 2018;39(33):3021–104.
- 12. Unger T, Borghi C, Charchar F, et al. 2020 International Society of Hypertension global hypertension practice guidelines. J Hypertens. 2020;38(6): 982–1004.
- 13. Markovitz AA, Mack JA, Nallamothu BK, Ayanian JZ, Ryan AM. Incremental effects of antihypertensive drugs: instrumental variable analysis. BMJ. 2017;359:j5542.
- 14. Paczkowska-Walendowska M, Sip S, Staszewski R, Cielecka-Piontek J. Single-pill combination to improve hypertension treatment: pharmaceutical industry development. Int J Environ Res Public Health. 2022;19:4156.
- 15. Ábrahám G, Dézsi CA. The antihypertensive efficacy of the triple fixed combination of perindopril, indapamide, and amlodipine: the results of the PETRA study. Adv Ther. 2017;34(7):1753–63.

16. Tsioufis K, Douma S, Kallistratos MS, Manolis AJ. Effectiveness and adherence to treatment with perindopril/indapamide/amlodipine single-pill combination in a Greek population with hypertension. Clin Drug Investig. 2019;39(4):385–93.